Author's response to reviews

Title: Inflammatory fibroid polyp of the ileum presenting with small bowel obstruction in an adult patient: a case report

Authors:

Toygar Toydemir (toygartoydemir@hotmail.com)

Version: 2 Date: 6 February 2010

Author's response to reviews:

1-The ethnicity of the patient is included in the case presentation section.
2-Declaration of written informed consent from the patient for publication of the manuscript and figures used, is included before the reference list.
3-The figure title section is included after the reference list.
4-A Competing interests and Authors’ contributions section is included before the reference list.